Transfer of thawed frozen embryo versus fresh embryo to improve the healthy baby rate in women undergoing IVF: the E-Freeze RCT

Abha Maheshwari, 1* Vasha Bari, 2 Jennifer L Bell, 2 Siladitya Bhattacharya, Priya Bhide, 3 Ursula Bowler,² Daniel Brison,⁴ Tim Child,⁵ Huey Yi Chong,¹ Ying Cheong,⁶ Christina Cole,² Arri Coomarasamy,7 Rachel Cutting,8 Fiona Goodgame,² Pollyanna Hardy,² Haitham Hamoda, Edmund Juszczak, 2,10 Yacoub Khalaf, 11 Andrew King, 2 Jennifer J Kurinczuk,² Stuart Lavery,¹² Clare Lewis-Jones, 13 Louise Linsell, 2 Nick Macklon, 14,15 Raj Mathur, 16 David Murray, 2 Jyotsna Pundir, 17 Nick Raine-Fenning, 18 Madhurima Rajkohwa, 19 Lynne Robinson, 20 Graham Scotland, 1 Kayleigh Stanbury 2 and Stephen Troup²¹ on behalf of the E-Freeze Trial **Collaborative Group**

¹Aberdeen Fertility Centre, NHS Grampian and University of Aberdeen, Aberdeen, UK

²Clinical Trials Unit National Perinatal Epidemiology Unit, Nuffield Department of Population Health, University of Oxford, Oxford, UK

³Assisted Conception Unit, Homerton University Hospital NHS Foundation Trust and Queen Mary University of London, London, UK

⁴Assisted Conception Unit, Manchester University NHS Foundation Trust, Manchester, UK

⁵Oxford Fertility, The Fertility Partnership, University of Oxford, Oxford, UK

⁶Complete Fertility Centre, University of Southampton, Southampton, UK

⁷Institute of Applied Health Research, University of Birmingham, Birmingham, UK

⁸Human Fertilisation and Embryology Authority, London, UK

⁹Assisted Conception Unit, King's College Hospital, London, UK

¹⁰Nottingham Clinical Trials Unit, University of Nottingham, Nottingham, UK

¹¹Assisted Conception Unit and Centre for Pre-implantation Genetic Diagnosis, Guy's and St Thomas' Hospital and King's College London, London, UK

- ¹²Assisted Conception Unit, Imperial College London, London, UK
- ¹³Fertility Network, London, UK
- ¹⁴London Women's Clinic, London, UK
- ¹⁵Gynaecology, University of Copenhagen, Copenhagen, Denmark
- ¹⁶Assisted Conception Unit, St Mary's Hospital, Manchester, UK
- ¹⁷Assisted Conception Unit, St Bartholomew's Hospital, London, UK
- ¹⁸The Fertility Partnership Nurture Fertility, Nottingham, UK
- ¹⁹CARE Fertility, Birmingham, UK
- ²⁰Gyanecology and Assisted Conception, Birmingham Women's and Children's NHS Foundation Trust, Birmingham, UK
- ²¹Reproductive Science Consultancy, Wilmslow, UK

Declared competing interests of authors: Vasha Bari, Pollyanna Hardy and Jennifer J Kurinczuk report receipt of funding from the National Institute for Health and Care Research (NIHR) during the conduct of the study. Jennifer L Bell, Ursula Bowler, Christina Cole, Fiona Goodgame, Andrew King, Louise Linsell, David Murray and Kayleigh Stanbury report receipt of funding from NIHR during the conduct of the study and outside the submitted work. Abha Maheshwari reports grants from NIHR and personal fees from Merck Serono (Darmstadt, Germany), Ferring Pharmaceuticals (Saint-Prex, Switzerland), Pharmasure Ltd (Watford, UK) and Cook Medical (Limerick, Ireland) outside the submitted work. Arri Coomarasamy reports that he is a member of the Efficacy and Mechanism Evaluation (EME) Funding Committee (2019–21). Pollyanna Hardy reports that she is a member of the Health and Technology Assessment (HTA) Commissioning Committee (2020-present). Siladitya Bhattacharya reports grants from the NIHR HTA programme during the conduct of the study and remuneration from Oxford University Press (Oxford, UK) for his role as Editor-in-Chief of Human Reproduction Open. Daniel Brison reports grants from NIHR during the conduct of the study, and grants from the European Commission (Brussels, Belgium), Diabetes UK (London, UK), NIHR, the European Society of Human Reproduction and Embryology (ESHRE) and the Medical Research Council (MRC) outside the submitted work. Ying Cheong reports personal fees from Merck Serono and Ferring Pharmaceuticals outside the submitted work. Edmund Juszczak reports receipt of funding from NIHR during the conduct of the study and outside the submitted work, membership of the HTA Commissioning Board (2013–16) and the NIHR HTA General Board (2016–17), and membership of the NHS England and NIHR partnership programme (2019-present). Edmund Juszczak is also Director of Clinical Trials Units funded by NIHR. Yacoub Khalaf reports receiving support for resources used to help with recruitment of participants in the study from Guy's and St Thomas' NHS Foundation Trust (London, UK) during the conduct of the study. Raj Mathur reports other private practice fees from Manchester Fertility (Cheadle, UK), and personal fees from Merck (Darmstadt, Germany), Ferring Pharmaceuticals and Gedeon Richter plc (Budapest, Hungary) outside the submitted work. Graham Scotland reports receiving travel expenses and accommodation to participate in an advisory board meeting on methodologies used and needed to evaluate the health economic value of assisted reproductive technologies treatment (consultancy) from Merck KGaA (Darmstadt, Germany) outside the submitted work. Stephen Troup reports personal fees from CooperSurgical International (Trumbull, CT, USA) and Parallabs Ltd (Watford, UK) outside the submitted work.

Published May 2022 DOI: 10.3310/AEFU1104

^{*}Corresponding author abha.maheshwari@abdn.ac.uk

Plain English summary

The E-Freeze RCT

Health Technology Assessment 2022; Vol. 26: No. 25

DOI: 10.3310/AEFU1104

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

During in vitro fertilisation, eggs and sperm are mixed in a laboratory to create embryos. An embryo is placed in the womb 2–5 days later (fresh-embryo transfer) and the remaining embryos are frozen for future use. Initial research suggested that freezing all embryos followed by thawing and replacing them a few weeks later could improve treatment safety and success. Although these data were promising, the data came from small studies and were not enough to change practice and policy.

We conducted a large, multicentre, clinical trial to evaluate the two strategies: fresh-embryo transfer compared with later transfer of frozen embryos. We also compared the costs of both strategies during in vitro fertilisation treatment, pregnancy and delivery.

This study was conducted across 18 clinics in the UK from 2016 to 2019, and 619 couples participated. Couples were allocated to one of two strategies: immediate fresh-embryo transfer or freezing of all embryos followed later by transfer of frozen embryo. The study's aim was to find out which type of embryo transfer gave participants a higher chance of having a healthy baby.

We found that freezing all embryos followed by frozen-embryo transfer did not lead to a higher chance of having a healthy baby. There were no differences between strategies in the number of live births, the miscarriage rate or the number of pregnancy complications. Fresh-embryo transfer was less costly from both a health-care and a patient perspective.

A routine strategy of freezing all embryos is not justified given that there was no increase in success rates but there were extra costs and delays to embryo transfer.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 13/115/82. The contractual start date was in August 2015. The draft report began editorial review in February 2021 and was accepted for publication in August 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2022 Maheshwari *et al.* This work was produced by Maheshwari *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk